Author’s response to reviews

Title: Article miRNA expression profiling in normal breast tissues and hereditary breast tumors from BRCA1- and BRCA2-germline mutation carriers in Brazil

Authors:

Danielle Pessôa-Pereira (dpessoap@outlook.com)
Adriane Evangelista (adriane.feijo@gmail.com)
Rhafaela Causin (rhafaela-lima@hotmail.com)
René da Costa Vieira (reneacv@terra.com.br)
Lucas Machado (lucasfariamachado@yahoo.com)
Iara Santana (iarasantana.hcb@gmail.com)
Vinicius Duval da Silva (vinids@gmail.com)
Karen Borba de Souza (kborbasouza@gmail.com)
Renato José de Oliveira-Silva (renatokjso@gmail.com)
Gabriela Fernandes (gabriela.carvalhofernandes@gamil.com)
Rui Reis (ruireis.hcb@gmail.com)
Edenir Palmero (edenirip@yahoo.com.br)
Marcia Marques (mmemsilveira@gmail.com)

Version: 3 Date: 13 Feb 2020

Author’s response to reviews:

RESPONSE LETTER

Technical Comments:

1. In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript. This information was provided in the last paragraph of the Funding section.

2. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case. This information was provided in the last paragraph of the Authors contributions section.